

Economic Research-Ekonomska Istraživanja

ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/rero20

Does share capital mater for company performance?

Tadeusz Dudycz

To cite this article: Tadeusz Dudycz (2022) Does share capital mater for company performance?, Economic Research-Ekonomska Istraživanja, 35:1, 3035-3059, DOI: 10.1080/1331677X.2021.1985575

To link to this article: https://doi.org/10.1080/1331677X.2021.1985575

© 2021 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group.



6

Published online: 04 Oct 2021.

ſ	
ļ	ک

Submit your article to this journal 🖸

Article views: 2296



View related articles

View Crossmark data 🗹

OPEN ACCESS

Francis Group

Routledge

Does share capital mater for company performance?

Tadeusz Dudycz 🗈

Faculty of Management, Wrocław University of Science and Technology, Wrocław, Poland

ABSTRACT

This article examines the impact of share capital on companies' performance as well as the effect of accounting information on companies' market performance and the impact of pre-IPO information on the predictive power of companies' performance after an initial public offering (IPO). The research was conducted on a sample of IPO companies debuting on the Warsaw Stock Exchange. It shows that a large percentage of share capital in equity reduces capital flexibility but can also be a signal to improve companies' market performance. It also shows that after an IPO, the market's information efficiency diminishes, which means, among other things, that pre-IPO accounting information has a negligible impact on the companies' market performance after the IPO.

ARTICLE HISTORY

Received 10 March 2021 Accepted 21 September 2021

KEYWORDS

Share capital; face value; capital flexibility: information efficiency; company performance; IPO

JEL CLASSIFICATIONS G14; G32; K22; M41; O16

1. Introduction

Capital structure' is commonly understood to mean the combination of the different types of financing sources that a company maintains as a result of its decisions to achieve specific benefits (Niu, 2008). The most prolific stream of research related to capital structure concerns the impact of debt on the value of an enterprise. This was initiated by the pioneering researchers Modigliani and Miller (1958), who claimed that, in the conditions of an excellent capital market, the amount of debt does not affect the value of a company. Their article contributed to the explosion of research on the structure of capital and its impact on a company's achievements. It led to the creation of many theories, among which trade-off theory, pecking order theory and agency theory are of particular importance. However, as Le and Phan (2017) note, there is no theory that fully explains the impact of debt on a company's performance. According to Ardalan (2017), this is due to the extraordinary complexity and diversity of societies, which is not taken into account by theories. Other aspects of capital structure, especially the impact of the equity structure on a company's performance, arouse much less interest in researchers. With the exception of the influence of ownership structure on a company's performance (Aluchna & Kaminski, 2017;

© 2021 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group.

CONTACT Tadeusz Dudycz 🖾 tadeusz.dudycz@pwr.edu.pl

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/ licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Kapopoulos & Lazaretou, 2007; Kaserer & Moldenhauer, 2008; Krivogorsky & Grudnitski, 2010; Maury, 2006; Short & Keasey, 1999), there is practically no research in this area. This is particularly true of the impact of share capital, which has special features that set it apart from other types of equity. Share capital is associated with the concept of par value, which has more than 200 years of history and was created to counteract fraudulent practices that became popular in the nineteenth century after shareholders were released from their obligations to the company (Cook, 1921). The concept of par value has been criticized since its birth, as it is said not to be able to protect creditors effectively against the defrauding behaviour of shareholders. Although it has become a cornerstone of the legislation of many countries, today there is a clear tendency to abandon it, except in the European Union and the European Economic Area. The strong position held by the concept of par value here results from its support in the Second Council Directive of 13 December 1976. Subsequent analyses, such as the Reforming capital report prepared by the Interdisciplinary Group on Capital Maintenance (Rickford, 2004) and the Feasibility study on an alternative to the capital company regime published by KPMG on behalf of the European Commission (KPMG, 2008), did not find unambiguous reasons for abandoning it. Therefore, it should be expected that this concept will continue to apply in the legislation of many countries.

The influence of share capital on a company's achievements may result from two important features. First, it is the capital that companies must hold to secure creditors' claims. A company cannot redistribute its share capital, and new shares may not be sold for less than par value. Its inviolable nature may therefore contribute to increasing the inflexibility of capital and thus negatively affect the company's performance. On the other hand, share capital can be a measure of shareholder involvement and thus be a signalling tool (Mulbert & Birke, 2002) for a company wishing to improve its image in the capital market.

The aim of this article is to examine whether these two characteristics of share capital have an impact on company performance. Additionally, the impact of pre-IPO accounting information on the predictive power of future company results as well as the impact of accounting information on the market performance of companies are examined.

In this article, the object of research is IPO firms whose debut took place in the period 1998–2013 on the Warsaw Stock Exchange (WSE). For each company, the results for a period of seven years were analysed, starting from one year before the IPO and ending five years after the IPO, which means that the analysis period covers the years 1998–2018.

The article contributes to the literature in several ways. First, it shows that, if share capital forms a large proportion of equity, it reduces capital flexibility and negatively affects accounting efficiency measures, such as the return on assets (ROA) and return on equity (ROE). On the other hand, it shows that there are reasons to think that, when share capital forms a large proportion of equity, it improves the financial image of the company and contributes to enhancing its market performance. Thus, the paper contributes to research on the impact of lender protection on various aspects of company operations. In the traditional approach the protection of lenders is

implemented through the principle of capital maintenance, to which the regime of par value and the resulting share capital belong. The par value regime is normally considered to be a technical issue for lawyers and accountants, but it can significantly affect the financing of a company (Rickford, 2004). In addition, research has suggested that creditor protection has an impact on companies' capital structure (Cho et al., 2014) and post-IPO operating performance (Espenlaub et al., 2020). Espenlaub et al. (2020) classified Poland as a country with low lender protection and showed that this protection has a negative impact on post-IPO operating performance. The research presented here sheds new light on the mechanism through which creditor protection influences post-IPO performance and on the multidirectional impact of share capital on firms' operating and market performance, which, to our knowledge, has not been reported in the literature so far.

Secondly, the findings show that it is possible to talk about market information efficiency and the impact of accounting information on market performance during the debut of a company and while prospectuses are being published. After the debut, the market performance breaks away from the fundamentals and is determined mainly by timing - the mood of investors. This is against the predictions of Ong et al. (2020) for the Malaysian market. Thus, the paper provides new knowledge about the mechanisms governing 'relational investor' markets, which constitute an area that has still not been fully explored, especially in emerging markets. The relationship between accounting information and market performance belongs to the most widely researched topics in accounting (Chen & Zhang, 2007). Most of the evidence, however, comes from the US market (Skogsvik, 2008), which represents a 'market-centred' model of governance (Pistor, 2000; Roe, 1993). According to Goslin et al. (2012), this does not warrant the conclusion that these phenomena are the same in all markets. The article shows a new phenomenon that has not previously been reported. Accounting information is assimilated by investors when it is accompanied by noise related to entering the stock exchange and is no longer assimilated when stocks become one of many securities.

Thirdly, research has shown that pre-IPO accounting information has very little predictive power for the long-term performance of companies (longer than one year). This phenomenon was already signalled by Bhabra and Pettway (2003), but the article provides evidence that the reasons behind it may be the low information efficiency of the market and the detachment of the market performance from the foundations in the years after the IPO. This is of great importance both for investors and for standard setters. The creation and dissemination of the International Financial Reporting Standards (IFRSs) is associated with high expectations regarding their impact on the functioning of capital markets. However, the work on and testing of new standards will be ineffective when the stock prices are disconnected from the fundamentals.

The remainder of the paper is organized as follows. Section 2 provides a literature review and the hypotheses for the investigation. Section 3 describes the data, the variables, and the research methods used in this study. Section 4 presents the empirical results, and Section 5 discusses the results and makes some concluding remarks.

2. Literature review and research hypotheses

2.1. Share capital

The concept of par value was conceived more than 200 years ago as a response to the fraudulent behaviour of shareholders after their release from personal responsibility for their company's obligations. The concept of par value consists of assigning each share a nominal value reflecting cash payments or assets contributed by each shareholder (Ho & Lan, 1999). Share capital can be considered the first legal capital because it initiated the principle of maintaining capital created to protect the company's creditors from the extra risk related to the limited liability of shareholders for the company's liabilities (Armour, 2000). Legal capital fulfils a similar role for creditors as a financial cushion which acts in the same way as financial adequacy provisions for financial institutions (Bachner, 2009; Handschin, 2012).

Legal capital has been criticized from its inception. Enriques and Macey (2001) believe that the capital-maintaining principle does not bring any benefits to creditors; on the contrary, in some cases it even harms them. Consequently, the accompanying burdens of this principle on companies and societies are completely unjustified. The authors are convinced that maintaining this principle in European law, despite its ineffectiveness, results only because of the influence of interest groups that benefit from the functioning of legal capital. These include incumbent management boards representing the interests of controlling shareholders, accounting officers providing required share valuation services, and lawyers assisting managers in navigating the maze of unnecessarily complicated laws regarding legal capital. In addition, Ho (2017) claims that the principle of capital maintenance contributes to reducing the flexibility of the financial structure and burdens it with cumbersome procedures that force companies to pay for 'useless expert reports and legal advice' (p. 19). Similarly, Mulbert and Birke (2002) consider that the utility of the capital maintaining principle for the protection of creditors is small, and there is a nontrivial likelihood that this principle does more harm than good. In turn, Mwenda (1999) remarks that the value of assets may increase or decrease over time and, for this reason, this value ceases to correspond to the value of the originally contributed share capital. Therefore, any cash-based equity valuations used as a signal for a market about the value of shares in the company's equity are usually fictional and may be both meaningless and misleading. Therefore, in contrast to Bebchuk (1992), Ho (2017) is convinced that in the real world, creditors do not pay attention to legally maintained capital, which should reflect the value of collateral, because they often protect their interests by covenants.

However, there are also authors who see the benefits of the capital-maintaining principle. Bonbright (1924) points out that par value does not reflect the value of a company subject to constant change and cannot be determined on the basis of share certificates; it only reflects the capital that shareholders agreed to contribute. In addition, Pennington (1990) emphasizes that the concept of par value is useful when declaring dividends as a percentage of par value, when determining the voting rights at a general meeting or determining the amounts due to privileged shareholders in the event of a company's liquidation. According to Mulbert and Birke (2002), the most important positive effect of the capital-maintenance principle is limiting harmful actions to creditors by distributing capital, in particular by way of 'hidden

distributions'. Therefore, the principle of maintaining capital can be a way of reducing post-contractual opportunism by shareholders (Armour, 2000), and the same mechanism can be used to reduce agency conflict (Panetsos, 2016). For example, as reported by Boubakri and Ghouma (2010), covenants may be ineffective in protecting creditors if the ultimate owners are a family. Cascino et al. (2013) are advocates of a continental accounting model which focuses on creditors and requires highly codified reporting (Joos & Lang, 1994), and empirical evidence from Europe shows that creditors prefer conservative accounting valuations. Mulbert and Birke (2002) states that the contribution of a significant amount of capital by shareholders sends out the signal that shareholders have confidence in their company and intend to work hard to succeed with the venture. We formulate the following hypothesis:

H1: Share capital is an effective signalling tool contributing to the improvement of companies' market performance.

2.2. Capital flexibility

'Financial flexibility is defined as the company's ability to take advantage of (cope with) a positive (negative) shock in its investment opportunity set' (Lambrinoudakis et al., 2019, p. 2). Anticipating future investment opportunities, managers try to prepare companies for the expected investment shock. Acting proactively, they reduce their debt so that in the future they will have a greater debt capacity. DeAngelo et al. (2011) justify this with a lower cost of debt than equity. In turn, collecting and keeping cash has a negative impact on value. Survey studies conducted by Graham and Harvey (2001), Bancel and Mittoo (2004) and Brounen et al. (2006) show that obtaining flexibility of capital is a very important factor in determining the financial policies of companies in the U.S and Europe. We assume that equity can also be a tool for achieving financial flexibility, either by allocating it to repayment of debt or by affecting creditworthiness. Therefore, if share capital of an inviolable nature has a large share in equity, the usefulness of equity as a tool in shaping financial flexibility decreases (Ho, 2017). This is suggested in the study by Bancel and Mittoo (2004), who showed that efforts to maintain financial flexibility are greater in countries with bank-based systems that use a continental accounting model focused mainly on valuing collateral for lenders. The results of Brounen et al. (2006) also show that financial flexibility is not driven by the peckingorder theory. In connection with the above, we formulate the following hypothesis:

H2: Share capital makes financial structure inflexible.

2.3. Predictive power of information from pre-IPO on post-IPO performance

An initial public offering is expected to accelerate a company's growth potential by raising sufficient capital (Mun & Jang, 2019). Investors analyse the information contained in prospectuses, estimate the growth potential, and make decisions to buy shares, which determine the success of the IPO. It can therefore be presumed that the information conveyed through a company's prospectus, which had an impact on the valuation of the new issue, is predictive of the future benefits for investors. The first research on the importance of accounting information in forecasting a company's future performance was initiated by Ou and Penman (1989), who showed that 'trading strategies based on predictions of future earnings from "publicly available" financial statement information capture a significant portion of returns'. Since then, the relationship between equity returns and accounting information has been one of the most widely researched topics in accounting (Chen & Zhang, 2007). According to Kothari (2001), the investment strategy that relies on financial statement analysis data forms a discrete field of research in accounting. There is a widely accepted view that the information contained in a financial statement is a valuable predictor of the future stock return. However, this view is mainly based on evidence from the US market (Amel-Zadeh et al., 2020; Bhabra & Pettway, 2003; Chen & Zhang, 2007; Danielson & Press, 2003; Frank, 2002; Kourtis et al., 2017; Piotroski, 2000; Yan & Zheng, 2017). It is the largest and one of the oldest markets with a 'market-centred' model of corporate governance, characterized by equity finance and control by capital markets (Pistor, 2000; Roe, 1993).

Regarding other markets, it was confirmed by Skogsvik (2008) and Skogsvik and Skogsvik (2010) for Sweden, by Alexakis et al. (2010) for Greece, by Chung and Kim (2001) for Korea, and by Goslin et al. (2012) for New Zealand. For emerging markets, with some exceptions (e.g., Nigeria; Ajekwe & Ibiamke, 2018), research in this area is very scarce. Therefore, as emphasized by Goslin et al. (2012), more evidence from different countries is needed to draw conclusions about the commonness of the relationship between accounting information and equity returns. It is all the more important as Woodley et al. (2011) indicated that this relationship may change over time.

Poland belongs to the 'relational investor' markets characterized by control through a coalition of banks, institutions, families, and shares between enterprises. Many companies are also owned by the state and state institutions. Such markets may have different investor behaviour, as Goslin et al. (2012) predicted for New Zealand. This prediction is supported by the research conducted by Morck et al. (2000), who showed that, in emerging markets such as Poland, stock exchanges are weak processors of economic information and share prices do not behave specifically. Some authors have also suggested that financial reporting in developing countries is of lower quality and less relevant to stock valuation than is the case in common-law countries (Ball, 2006; Ball et al., 2000). In addition, almost all of the studies mentioned focused on the short-term relationship (one year), while single studies (Bhabra & Pettway, 2003) have not confirmed this relationship in the long term. We assume that if the current accounting information affects the valuation of new-issue shares, it means that, for investors, the basis for their valuation is the fundamental value, which is realized in the long term. Therefore, we formulate the following hypothesis:

H3: Information affecting the valuation of new issues is a good predictor of a company's future successes.

3. Sample description and research methods

3.1. Sample

The study is based on a sample of IPO firms which went public on the Warsaw Stock Exchange from 1998 to 2013 and five subsequent years after IPO, so the

analysis cover the period 1998–2018 (total number of IPOs reported in Panel A of Table 1). This initial sample of 496 IPOs was reduced by excluding:

- a. financial institutions (e.g., banks and insurance companies);
- b. IPOs which were not connected with new common stock issuance; and
- c. IPOs for which data was incomplete.

The final sample thus consists of 259 IPOs. Panel A of Table 1 shows the variations in the number of IPOs in the sample over the 16 years of the study.

The primary source of data used in this study was Notoria Service; however, because there are sometimes incomplete or incorrect data in the database, other sources of data were used to supplement these, such as IPO prospectuses and annual reports available on companies' websites and at www.gpwinfostrefa.pl. Market data about the companies comes directly from the WSE website: www.gpw.pl. Panel B of Table 1 reports the basic characteristics of the sample firms one year before their IPOs, such as total assets and total sales.

3.2. Research methods

The study was carried out using multiple regression analysis, which is a standard method for estimating the relationship between the determinants of financial performance and financial performance itself (Paniagua et al., 2018). Two types of models were built. The first type examined the relationship of independent variables from a given year to a company's performance in the same year. In the second type of model used to study predictive power, the relationship of independent variables measured one year before IPO to company performance after the IPO was investigated. To measure the impact of share capital on the inflexibility of capital, the accounting measures of performance were used as the dependent variables. Accounting measures are directly linked to decisions taken in the company; therefore, the effects of decisions aimed at adjusting the amount of capital to current needs are quickly visible in accounting measures. We estimated the following equation:

$$I_{A}: Y_{t} = a + \beta_{1}ROS_{t} + \beta_{2}\frac{S}{A_{t}} + \beta_{3}CR_{t} + \beta_{4}\frac{D}{A_{t}} + \beta_{5}\frac{E}{FA_{t}} + \beta_{6}\frac{SC}{E_{t}} + \beta_{7}Ln \ Assets_{t}$$
$$+ \beta_{8}Equity \ growth_{t} + \beta_{9}Sector + \beta_{10}WIG_{t} + \beta_{11}GDP[\%]_{t} + \varepsilon_{t}$$

$$\begin{aligned} \text{II}_{\text{A}} : Y_t &= a + \beta_1 ROS_{IPO-1} + \beta_2 \frac{S}{A_{IPO-1}} + \beta_3 CR_{IPO-1} + \beta_4 \frac{D}{A_{IPO-1}} + \beta_5 \frac{E}{FA_{IPO-1}} \\ &+ \beta_6 \frac{SC}{E_{IPO-1}} + \beta_7 Ln \ Assets_{IPO-1} + \beta_8 Equity \ growth_{IPO-1} + \beta_9 Sector \\ &+ \beta_{10} WIG_{IPO-1} + \beta_{11} GDP[\%]_{IPO-1} + \varepsilon_{IPO-1} \end{aligned}$$

where the dependent variable Y_t is ROA or ROE, t is the year relative to IPO (IPO + 0 means the year of debut, IPO-1 is one year before debut, IPO + 1 is the first year

acteristics.	
ole char	
1. Samp	
Table '	

		1998	1999	2000	2001	2002	2003	2004	2005	2006	2007	2008	2009	2010	2011	2012	2013	1998-2013
Panel A																		
Total number of IPOs		51	20	11	7	S	9	36	58	63	81	31	13	34	38	19	23	496
Sample number of IPOs		25	15	6	S	2	S	24	22	25	52	22	6	21	12	4	7	259
Sample number of IPOs to		49.0	75.0	81.8	71.4	40.0	83.3	66.7	37.9	39.7	64.2	71.0	69.2	61.8	31.6	21.1	30.4	52.2
total number of IPOs [%	[
GDP growth [%]		5.0	4.5	4.3	1.2	1.4	3.9	5.3	3.6	6.2	6.8	5.1	1.6	3.8	4.3	1.9	1.6	74.5
MIG		12,795.6	18,083.6	17,847.6	13,922.2	14,366.7	20,820.1	26,636.2	35,600.8	50,411.8	55,648.5	27,228.6	39,986.0	17,489.9	37,595.4	47,460.6	1,284.3	
Panel B																		
Assets $_{-1}$ [PLN million]	Mean	44.8	142.5	596.2	43.1	86.1	134.9	130.5	1 044.6	118.9	97.9	496.1	2,719.5	129.7	115.2	312.9	183.3	331.9
	SD	35.9	172.4	1 158.3	33.8	52.8	57.0	212.7	3 134.6	168.5	152.6	1,868.1	7,160.0	146.2	85.6	253.5	123.5	1,736.8
	Median	34.8	52.5	39.4	41.7	86.1	122.0	74.2	52.4	55.9	52.1	29.0	69.9	93.9	96.3	329.3	161.6	54.7
Sales $_{-1}$ [PLN million]	Mean	91.4	235.8	174.0	47.5	269.4	162.8	107.3	966.3	144.5	118.3	343.6	1,243.6	90.6	98.1	144.0	82.0	253.5
	SD	101.9	260.9	249.5	67.7	299.5	85.1	221.2	2,215.7	256.9	130.9	1,072.9	3,190.0	114.5	88.5	156.8	141.8	961.8
	Median	52.8	78.5	63.1	23.1	269.4	189.3	49.2	7.7.7	50.0	62.5	26.8	36.5	53.7	69.2	120.7	8.0	54.9
F																		

Source: The Author.

3042 🕢 T. DUDYCZ

after debut. IPO + 5 is the fifth year after debut), and ε is a stochastic error term. All other variables are defined in Table 2.

However, the following market performance measures were used to test the utility of share capital as a signalling tool – market value to book value (MV/BV) and Tobin's Q ratio (QT) and market profitability – Cumulative abnormal return (CAR) and Buy-and-hold abnormal return (BHAR) (Espenlaub et al., 2020). Market measures reflect investors' perceptions and show how they respond to specific signals. We estimated the following equation:

$$\begin{split} I_{M} : Y_{t} &= a + \beta_{1}ROE_{t} + \beta_{2}\frac{S}{A_{t}} + \beta_{3}CR_{t} + \beta_{4}\frac{D}{A_{t}} + \beta_{5}\frac{E}{FA_{t}} + \beta_{6}\frac{SC}{E_{t}} + \beta_{7}Ln \ S_{t} \\ &+ \beta_{8}NrSh_{t} + \beta_{9}\frac{IP}{NP} + \beta_{10}Sector + \beta_{11}WIG_{t} + \beta_{12}GDP[\%]_{t} + \varepsilon_{t} \\ II_{M} : Y_{t} &= a + \beta_{1}ROE_{IPO-1} + \beta_{2}\frac{S}{A_{IPO-1}} + \beta_{3}CR_{IPO-1} + \beta_{4}\frac{D}{A_{IPO-1}} + \beta_{5}\frac{E}{FA_{IPO-1}} \\ &+ \beta_{6}\frac{SC}{E_{IPO-1}} + \beta_{7}Ln \ S_{IPO-1} + \beta_{8}NrSh_{IPO-1} + \beta_{9}\frac{IP}{NP} + \beta_{10}Sector \\ &+ \beta_{11}WIG_{IPO-1} + \beta_{12}GDP[\%]_{IPO-1} + \varepsilon_{IPO-1} \end{split}$$

where the dependent variable are MV/B, Q-T, CAR, BHAR, while t and ε are analogically as above.

As an independent variable, the share of share capital in equity was used. For other control variables, we used both financial indicators showing the financial situation of companies and parameters reflecting the state of the economy (GDP) and the mood of investors (WIG – Warsaw Stock Index) and the sector. To a small extent, these financial indicators were determined by the specifics of the sector, and therefore they have a high comparability value. Those parameters that had a strongly asymmetrical distribution were used in logarithmic form. The variables used in this study are defined in Table 2.

4. Results

4.1. The relationship of share capital to accounting performance

In Table 3 Panel B1, we see that SC/E has a negative effect on ROA. It occurs in the year before the IPO, the year of issue, and in the third and fourth years after issue. This is probably due to the inviolable nature of the SC which means that a company cannot fully adapt capital to current needs, especially when capital needs to be reduced. ROA is scaled with total capital, which means that capital inflexibility caused by SC's large share of equity has negative effects on profitability visible at the level of total capital. The impact of SC on capital flexibility at equity level is obviously stronger as seen in Table 4 Panel B1, where the statistically significant negative impact of SC/E on ROE has already occurred in six years. The negative impact of SC/E on profitability seems to support hypothesis 2 but with some limitations. The analysed sample of debuting companies recorded decreases in profitability after their debuts (Tables 3 and 4, Panel A), which is quite a common phenomenon (Auret & Britten, 2008; Cai & Wei, 1997; Jain & Kini, 1994; Kim et al., 2004; Mikkelson et al., 1997;

Table 2. Variable definitions	Table 2	2. Va	iriable	definitions
-------------------------------	---------	-------	---------	-------------

Variable	Definition
Dependent variables	
ROA	Return on assets defined as earnings before interests and taxes divided by total assets
ROE	Return on equity defined as net income divided by shareholders' equity, calculated as the difference between total assets and total liabilities
IP/BV	Issue price to book value of shares
MV/BV	The ratio of market value to the book value of shares
Q-T	Tobin's Q ratio approximated by using Chung and Pruitt (1994) formula: Approximate Q-T = (MVE + PS + D)/TA, where MVE is the product of a firm's share price and the number of common stock shares outstanding, PS is the liquidating value of the firm's outstanding preferred stock, D is the value of short-term liabilities net of short-term assets, plus the book value of the long-term debt, and TA is book value of the total assets.
CAR	Cumulative abnormal return calculated as: $CAR_T = \sum_{t=0}^{T} \left \frac{\sum_{i=1}^{t} (n_{it}^i - n_{it})}{N} \right $
BHAR	Buy-and-hold abnormal return calculated as:
	$BHAR_{T} = \frac{1}{N} \sum_{i=1}^{N} \left[\prod_{t=0}^{T} (1 + R_{it}) - \prod_{t=t_{i}}^{T} (1 + R_{bt}) \right]$ where R ^{it} is the return on IPO firm i in month t, R ^{Bt} is the return on the benchmark portfolio in month t, N is the number of IPO firms, and T is the number of months
Independent variables	or months
ROE	Return on equity
ROS	Return on sales ratio defined as earnings before interests and taxes divided by net sales
S/A	Net sales to total assets ratio
CR	Cash ratio defined as short-term investments divided by current liabilities
E/FA	Shareholders' equity-to-fixed-assets ratio
D/A	Debt ratio defined as total debt (the sum of current liabilities and long-term liabilities) divided by total assets
SC/E	Share capital to the total shareholders' equity ratio
Ln Assets	Natural logarithm of total assets
Ln S	Natural logarithm of net sales
Ln NrSh	Natural logarithm of number of shares issued
IP/NP	Isue price to nominal price
Equity growth	Percentage increase in shareholders' equity calculated as proceeds divided by sharelolders' equity before new issue
Sector	Sector of the economy. A variable used to control for the sector effect, which is coded as: 1 – architecture; 2 – chemical industry; 3 – energy; 4 – mining and metallurgy; 5 – trade; 6 – information technology; 7 – media and telecommunications; 8 – heavy industry; 9 – light industry; 10 – food industry; 11 – services.
WIG	Warsaw Stock Exchange Index
GDP [%]	GDP growth rate defined as the percentage change in gross domestic product during one year

Source: The Author.

Pagano et al., 1998). However, it should be assumed that in such a situation, there is a need to reduce capital rather than increase it. The impact of share capital on financial flexibility in the event of a positive investment shock and the need to increase capital require separate studies.

4.2. Usefulness of share capital as a signalling tool

Share capital can be considered a useful signalling tool if its increase in the equity structure is positively received by investors and improves their perceptions of the financial image of the company (Spence, 1973, 2002). To be able to infer this aspect of share capital, one must refer to other phenomena that can be observed. In Table 5 Panel B1, we see that SC/E has a strong and positive impact on the valuation of the

Table 3. Regress	ion analysi	s of ROA.											
	ROA _{IPO-1}	ROA IPO+0	ROA IPO+1	ROA IPO+2	ROA IPO+3	ROA IPO+4	ROA IPO+5	ROA IPO+0	ROA IPO+1	ROA IPO+2	ROA IPO+3	ROA IPO+4	ROA IPO+5
Panel A: Mean value													
Z	259	259	259	259	252	233	215						
Mean	0.127	0.078	0.042	-0.002	0.021	0.029	0.041						
Trimmed mean 5%	0.122	0.080	0.052	0.027	0.025	0.030	0.024						
SD	0.124	0.094	0.120	0.246	0.244	0.187	0.422						
Median	0.104	0.073	0.052	0.033	0.032	0.031	0.031						
Panel B: Regression a	nalysis												
	Panel B1: In	dependent var.	iables from ec	ach year				Panel B2: Inc	dependent va	riables from	the year bef	ore IPO	
ROS	0.004	0.048	0.050	0.014	0.303***	0.387***	0.214***	0.004	-0.052	-0.011	-0.012	-0.019	-0.034
	(0.959)	(0.436)	(0.402)	(0.780)	(0000)	(0000)	(0.001)	(0.959)	(0.470)	(0.886)	(0.866)	(0.795)	(0.665)
S/A	060.0	0.118**	0.092	0.063	0.126**	0.102*	0.372***	060.0	0.121*	0.036	0.017	0.042	-0.013
	(0.132)	(0.047)	(0.125)	(0.197)	(0.013)	(060.0)	(0000)	(0.132)	(0.052)	(0.575)	(0.791)	(0.516)	(0.846)
CR	-0.123^{*}	-0.189***	-0.112	-0.119^{**}	-0.058	-0.011	0.020	-0.123^{*}	-0.236^{***}	-0.157^{**}	-0.026	0.004	-0.012
	(0.092)	(0.004)	(0.058)	(0.015)	(0.248)	(0.859)	(0.757)	(0.092)	(0.002)	(0.048)	(0.744)	(0.964)	(0.888)
D/A	-0.043	-0.159**	-0.424***	-0.658***	-0.446***	-0.358***	-0.028	-0.043	0.115*	-0.037	-0.152**	-0.066	0.044
	(0.468)	(0.012)	(000.0)	(000.0)	(0000)	(0000)	(0.677)	(0.468)	(0.079)	(0.591)	(0.026)	(0.338)	(0.546)
E/FA	0.059	0.014	0.060	-0.113^{**}	0.001	-0.063	0.038	0.059	0.064	0.076	0.050	-0.011	0.011
	(0.366)	(0.823)	(0.299)	(0.038)	(0.978)	(0.362)	(0.644)	(0.366)	(0.348)	(0.281)	(0.476)	(0.882)	(0.888)
SC/E	-0.149^{**}	-0.283***	-0.085	-0.074	-0.116^{**}	-0.117**	-0.045	-0.149^{**}	-0.116^{*}	-0.048	0.019	-0.104	-0.048
	(0.016)	(0000)	(0.152)	(0.163)	(0.020)	(0.040)	(0.589)	(0.016)	(0.084)	(0.488)	(0.788)	(0.137)	(0.509)
Ln Assets	-0.331***	0.027	0.075	0.187***	0.214***	-0.015	-0.047	-0.331***	-0.064	-0.022	0.103	-0.017	-0.111
	(0000)	(0.676)	(0.225)	(000.0)	(0000)	(0.807)	(0.502)	(000.0)	(0.367)	(0.767)	(0.165)	(0.823)	(0.160)
Equity growth [%]		-0.124^{*}	-0.043	-0.014	0.034	-0.059	-0.126^{*}		-0.124	-0.005	0.082	0.078	-0.058
		(0.059)	(0.486)	(0.779)	(0.507)	(0.318)	(0.061)		(0.108)	(0.947)	(0.301)	(0.337)	(0.487)
Sector	0.005	0.025	-0.032	-0.015	-0.017	0.068	0.029	0.005	-0.004	-0.068	-0.075	0.098	0.072
	(0.934)	(0.678)	(0.586)	(0.758)	(0.730)	(0.257)	(0.660)	(0.934)	(0.953)	(0.297)	(0.244)	(0.133)	(0.303)
DIM	-0.033	-0.087	-0.016	0.006	0.132**	0.106*	0.165**	-0.033	0.043	0.049	-0.018	0.147**	0.063
	(0.579)	(0.188)	(0.795)	(006.0)	(0.018)	(0.074)	(0.018)	(0.579)	(0.508)	(0.470)	(0.793)	(0.032)	(0.372)
GDP [%]	0.053	0.195***	0.075	0.055	-0.039	0.059	0.073	0.053	0.084	-0.052	-0.130^{**}	-0.062	-0.098
	(0.380)	(0.003)	(0.235)	(0.263)	(0.469)	(0.303)	(0.287)	(0.380)	(0.181)	(0.425)	(0.046)	(0.350)	(0.155)
\mathbb{R}^2	0.167	0.180	0.197	0.450	0.435	0.314	0.220	0.167	0.108	0.035	0.045	0.048	0.029
щ	4.980	4.920	5.511	18.347	16.617	9.044	5.088	4.980	2.732	0.812	1.063	1.110	0.610
Panel A reports the	mean values	of each varia	ble Yt. IPO-1	to IPO + 5 de	anote the yea	r in relation t	o IPO, where	IPO + 0 is the theorem of theorem of theorem of the theorem of t	he year of IP(0			
The results of the re	gressions fro	im the model	I _A are reporte	ed in Panel B1	and from th	e model II _A ir	n Panel B2.		1				,
The constant term i	s not report	ed. <i>P</i> -values a	re reported i	n parentheses	: (*), (**) and	d (***) indici	ate that coeff.	icients are si	gnificant at t	the 10, 5 an	d 1 percent	levels of sig	gnificance,
respectively. Source: The Author.													

Table 4. Regressi	on analysis	of ROE.											
	ROE _{IPO-1}	ROE IPO+0	ROE IPO+1	ROE IPO+2	ROE IPO+3	ROE IPO+4	ROE IPO+5	ROE IPO+0	ROE IPO+1	ROE IPO+2	ROE IPO+3	ROE IPO+4	ROE IPO+5
Panel A: Mean value													
Z	259	259	257	249	241	224	207						
Mean	0.219	0.092	0.020	-0.076	-0.134	-0.098	-0.413						
Trimmed mean 5%	0.214	0.101	0.063	0.020	-0.004	0.038	0.019						
SD	0.238	0.149	0.432	0.722	1.113	1.450	5.536						
Median	0.196	0.100	0.071	0.045	0.043	0.046	0.044						
Panel B: Regression a	nalysis												
I	Panel B1: Ina	lependent var	iables from ea	ich year				Panel B2: Ina	lependent var	riables from th	he year befor	e IPO	
ROS	0.033	0.331***	0.863***	-0.119^{***}	0.561***	0.009	0.320***	0.033	-0.012	0.160***	0.030	0.002	-0.007
	(0.625)	(0000)	(0000)	(0.008)	(0000)	(0.770)	(0000)	(0.625)	(0.855)	(0000)	(0690)	(0.979)	(0.934)
S/A	0.078	0.107	0.041	0.034	0.079	0.033	0.186***	0.078	0.122**	0.051	0.029	0.033	0.016
	(0.188)	(0.057)	(0.083)	(0.444)	(060.0)	(0.302)	(0.007)	(0.188)	(0.034)	(0.144)	(0.652)	(0.627)	(0.820)
ß	-0.150^{**}	-0.122^{**}	-0.005	-0.051	-0.089	-0.053	0.018	-0.150^{**}	-0.446^{***}	-0.799^{***}	-0.104	-0.015	0.004
	(0.038)	(0.046)	(0.832)	(0.259)	(0.054)	(0.115)	(0.792)	(0.038)	(000.0)	(0000)	(0.189)	(0.860)	(0.958)
D/A	0.187***	0.057	-0.052^{**}	-0.221^{***}	-0.195^{***}	-0.051	-0.207^{***}	0.187***	0.152**	-0.148^{***}	-0.204^{***}	-0.025	-0.098
	(0.002)	(0.338)	(0.033)	(000.0)	(0000)	(0.115)	(0.003)	(0.002)	(0.013)	(0000)	(0.003)	(0.730)	(0.192)
E/FA	0.085	0.022	0.019	0.011	0.005	0.015	0.044	0.085	0.114	0.282***	0.076	0.046	-0.008
	(0.187)	(0.695)	(0.417)	(0.800)	(0.904)	(0.621)	(0.487)	(0.187)	(0.069)	(0000)	(0.281)	(0.530)	(0.914)
SC/E	-0.166^{***}	-0.274***	-0.159^{***}	-0.656***	-0.363^{***}	-0.895***	-0.037	-0.166^{***}	-0.077	0.011	-0.056	0.010	-0.035
	(0.006)	(0000)	(0000)	(000.0)	(0000)	(0000)	(0.578)	(0.006)	(0.216)	(0.763)	(0.423)	(0.886)	(0.636)
Ln Assets	-0.255***	0.092	0.045	0.073	0.087	-0.024	0.204***	-0.255***	0.018	0.022	0.104	0.092	0.039
	(0000)	(0.130)	(0.066)	(0.120)	(0.071)	(0.450)	(0.003)	(0000)	(0.787)	(0.542)	(0.155)	(0.234)	(0:630)
Equity growth [%]		-0.058	-0.025	-0.040	-0.007	-0.006	-0.146^{**}		-0.072	0.001	0.115	0.061	0.083
		(0.349)	(0.318)	(0.376)	(0.878)	(0.841)	(0:030)		(0.298)	(0.980)	(0.142)	(0.465)	(0.323)
Sector	0.008	0.021	0.002	0.023	-0.053	0.068**	0.066	0.008	-0.006	-0.035	-0.131**	-0.028	0.077
	(0.894)	(0.710)	(0.940)	(0.607)	(0.250)	(0.028)	(0.312)	(0.894)	(0.914)	(0.315)	(0.046)	(0.683)	(0.291)
DIM	0.084	0.013	0.060**	-0.021	-0.040	-0.023	-0.055	0.084	0.077	0.073**	-0.029	-0.068	0.071
	(0.153)	(0.840)	(0.015)	(0.653)	(0.424)	(0.454)	(0.417)	(0.153)	(0.188)	(0.039)	(0.656)	(0.322)	(0.326)
GDP [%]	0.052	0.124**	0.015	0.084	0.050	-0.001	-0.071	0.052	0.000	-0.144***	-0.100	0.023	-0.053
	(0.378)	(0.045)	(0.547)	(0.064)	(0.301)	(0.978)	(0.291)	(0.378)	(0.995)	(0000)	(0.125)	(0.734)	(0.457)
R ²	0.187	0.260	0.875	0.559	0.554	0.826	0.269	0.187	0.237	0.720	0.0.72	0.014	0.024
ш	5.690	7.880	155.71	27.296	25.685	90.057	6.368	5.690	6.967	57.334	1.671	0.288	0.468
As in Table 3. Source: The Author.													

3046 🍚 T. DUDYCZ

Table 5. Regr	ession anal	ysis of the I	market to b	ook value i	ratio.								
	IP/BV _{IPO-1}	MV/BV _{PO+0}	MV/BV _{IPO+1}	MV/BV _{IPO+2}	MV/BV _{IPO+3}	MV/BV _{IPO+4}	MV/BV _{PO+5}	MV/BV _{IPO+0}	MV/BV _{IPO+1}	MV/BV _{PO+2}	MV/BV _{IPO+3}	MV/BV _{IPO+4}	MV/BV _{PO+5}
Panel A: Mean value													
z	259	259	259	259	252	233	214						
Mean	5.964	2.445	1.387	1.830	2.103	1.323	1.882						
Trimmed mean 5%	4.806	2.223	1.766	1.671	1.450	1.188	1.247						
SD	7.744	1.923	10.810	2.734	9.632	2.933	5.805						
Median	3.715	1.938	1.366	1.322	1.194	0.944	1.038						
Panel B: Regression and	alysis												
•	Panel B1: Indep	rendent variables fi	rom each year					Panel B2: Indep	endent variables fi	rom the year befo	re IPO		
ROE	0.290***	0.241***	0.092	-0.274***	-1.059^{***}	-0.180	-0.179^{**}	0.236***	0.018	0.036	-0.078	0.189***	0.019
	(0000)	(0000)	(0.187)	(0.002)	(0000)	(0.141)	(0.015)	(0000)	(0.793)	(0.598)	(0.267)	(0.008)	(0.809)
S/A	0.013	0.004	0.013	0.015	0.041	-0.007	-0.010	-0.018	-0.029	-0.018	0.033	-0.059	-0.001
	(0.796)	(0:950)	(0.832)	(0.806)	(0.104)	(0.912)	(0.904)	(0.738)	(0.654)	(0.776)	(0.619)	(0.373)	(0.986)
GR	0.036	-0.045	-0.101	-0.033	0.004	-0.043	-0.029	0.100*	0.061	-0.001	-0.041	0.067	-0.047
	(0.499)	(0.471)	(0.120)	(0.629)	(0.860)	(0.453)	(0.691)	(0.092)	(0.385)	(0.988)	(0.568)	(0.357)	(0.557)
D/A	0.251***	0.147**	0.160**	0.121*	-0.051^{*}	0.136**	0.157**	0.129**	0.073	0.143**	0.097	0.087	0.137*
	(000.0)	(0.015)	(0.016)	(0.079)	(0.052)	(0.018)	(0.042)	(0.028)	(0.295)	(0.039)	(0.168)	(0.223)	(0.078)
E/FA	-0.034	0.017	-0.014	-0.053	-0.003	-0.011	-0.021	-0.091	-0.013	-0.028	-0.010	-0.024	0.003
	(0.514)	(0.768)	(0.815)	(0.368)	(0.890)	(0.833)	(0.754)	(0.114)	(0.850)	(0.680)	(0.882)	(0.736)	(0.965)
SC/E	0.428***	0.029	0.009	-0.196**	-0.396***	0.465***	-0.070	0.279***	0.134**	0.172**	-0.034	0.240***	-0.014
	(000.0)	(0.657)	(0.899)	(0.031)	(0000)	(0000)	(0.346)	(0000)	(0.049)	(0.011)	(0.621)	(0.001)	(0.855)
Ln S	-0.138**	-0.206***	-0.191**	-0.074	0.031	0.030	-0.048	-0.020	0.130*	0.002	-0.180**	0.136*	-0.162*
	(0.017)	(0.005)	(0.021)	(0.355)	(0.295)	(0.660)	(0.618)	(0.747)	(0.084)	(0.974)	(0.019)	(0.085)	(0.062)
Ln NrSh	-0.105**	-0.044	-0.086	-0.055	0.040	-0.120**	-0.107	-0.134^{**}	-0.112	-0.042	-0.009	-0.114	0.015
	(0.049)	(0.517)	(0.234)	(0.427)	(0.138)	(0.049)	(0.172)	(0.023)	(0.111)	(0.541)	(006.0)	(0.116)	(0.854)
IP/NP	0.190***	0.199***	0.088	0.050	-0.008	-0.025	-0.015	0.217***	0.075	0.113*	0.069	-0.019	0.021
	(000.0)	(0.001)	(0.144)	(0.407)	(0.724)	(0.634)	(0.824)	(0000)	(0.260)	(0.088)	(0.313)	(0.785)	(0.781)
Sector	-0.036	0.057	0.030	0.084	0.044*	0.059	0.100	0.058*	0.070	0.122*	0.035	0.016	-0.011
	(0.461)	(0.318)	(0.621)	(0.166)	(0.057)	(0.275)	(0.155)	(0.282)	(0.275)	(0.055)	(0.591)	(0.806)	(0.878)
DIM	0.217***	0.280***	0.373***	0.258***	0.054**	0.143***	0.239***	0.284***	-0.036	-0.105	0.032	-0.106	-0.122
	(000.0)	(000.0)	(0000)	(0000)	(0.039)	(600.0)	(0.002)	(0000)	(0.601)	(0.123)	(0.651)	(0.127)	(0.108)
GDP [%]	0.057	0.054	0.093	0.229***	0.023	0.062	0.203***	-0.113^{**}	0.070	-0.024	-0.086	-0.037	-0.068
	(0.248)	(0.375)	(0.126)	(0.000)	(0.351)	(0.228)	(0.004)	(0:039)	(0.282)	(0.705)	(0.193)	(0.571)	(0.346)
R ²	0.451	0.252	0.215	0.215	0.888	0.468	0.163	0.337	0.059	0.081	0.049	0.117	0.043
ш	16.821	6.890	5.560	5.450	149.390	15.385	3.111	10.442	1.292	1.809	1.032	2.427	0.739
Panel A reports t	he mean valu	ues of each ve	ariable Yt. IPO	-1 to IPO + $\frac{1}{2}$	5 denote the	vear in relation	on to IPO, wh	nere IPO + 0	is the vear of	f IPO.			
The results of the	e regressions	from the mod	del I _M are rep	orted in Pane	B1 and fror	n the model I	II _M in Panel B	32.					

The constant term is not reported. P-values are reported in parentheses. (*), (**) and (***) indicate that coefficients are significant at the 10, 5 and 1 percent levels of significance, respectively. Source: The Author.

ECONOMIC RESEARCH-EKONOMSKA ISTRAŽIVANJA 🍥 3047

Table 6. Regressiv	on analysis	of the Tobir	n's Q ratio.									
	Q-T 100+0	Q-T IPO+1	Q-T IPO+2	Q-T IPO+3	Q-T IPO+4	Q-T IPO+5	Q-T _{PO+0}	Q-T IPO+1	Q-T _{PO+2}	Q-T _{PO+3}	Q-T IPO+4	Q-T IPO+5
Panel A: Mean value												
Z	259	258	258	251	230	212						
Mean	1.293	1.119	1.166	0.962	1.010	1.107						
Trimmed mean 5%	1.139	0.927	0.951	0.843	0.732	0.793						
SD	1.345	1.380	1.677	3.271	2.247	1.932						
Median	0.998	0.776	0.790	0.669	0.557	0.633						
Panel B: Regression ai	nalysis											
•	Panel B1: Ind	lependent varia	ables from each	year			Panel B2: Ind	ependent varic	ibles from the	year before l	РО	
ROE	0.165**	0.050	-0.005	-0.372***	0.164	-0.100	0.218***	-0.001	0.019	-0.045	-0.078	-0.042
	(0.012)	(0.472)	(0.957)	(0000)	(0.309)	(0.170)	(0000)	(0.986)	(0.777)	(0.523)	(0.279)	(0.578)
S/A	0.011	0.055	0:030	0.064	-0.096	-0.091	-0.025	0.003	0.020	0.022	0.040	0.015
	(0.853)	(0.362)	(0.633)	(0.346)	(0.243)	(0.281)	(0.642)	(0.958)	(0.757)	(0.734)	(0.555)	(0.832)
CR	-0.105*	-0.153^{**}	-0.038	-0.027	-0.100	-0.101	0.068	-0.062	-0.100	-0.077	-0.101	-0.102
	(0.096)	(0.017)	(0.581)	(0.676)	(0.181)	(0.166)	(0.241)	(0.352)	(0.150)	(0.286)	(0.171)	(0.187)
D/A	-0.154^{**}	-0.066	-0.114	-0.097	0.072	0.057	-0.117^{**}	-0.059	0.008	-0.032	0.020	0.137*
	(0.011)	(0.307)	(0.103)	(0.170)	(0.335)	(0.453)	(0.043)	(0.377)	(906.0)	(0.647)	(0.780)	(0.071)
E/FA	-0.070	-0.110^{*}	-0.131^{**}	-0.113^{*}	-0.155^{**}	-0.084	-0.122^{**}	-0.079	-0.052	0.002	-0.029	-0.018
	(0.228)	(0.056)	(0.028)	(0.062)	(0.025)	(0.208)	(0.031)	(0.225)	(0.439)	(0.977)	(0.687)	(0.812)
SC/E	0.083	-0.009	0.017	-0.147^{**}	0.166	0.002	0.374***	0.204***	0.039	0.033	-0.094	-0.072
	(0.212)	(0.901)	(0.854)	(0.033)	(0.304)	(0.978)	(000.0)	(0.002)	(0.563)	(0.631)	(0.206)	(0.356)
Ln S	-0.199***	-0.231***	-0.083	-0.021	0.034	0.019	-0.001	-0.137^{*}	-0.198***	-0.101	-0.211^{***}	-0.230^{***}
	(0.006)	(0.005)	(0.303)	(0.790)	(0.701)	(0.847)	(0.983)	(0.056)	(0.008)	(0.184)	(600.0)	(0.007)
Ln NrSh	-0.035	-0.024	-0.034	-0.025	-0.128	-0.150^{*}	-0.116^{**}	0.034	0.113	0.066	0.072	0.043
	(0.607)	(0.735)	(0.627)	(0.730)	(0.104)	(0.056)	(0.045)	(0.613)	(0.104)	(0.350)	(0.330)	(0.579)
IP/NP	0.252***	0.140**	0.107*	0.012	-0.047	-0.013	0.253***	0.220***	0.079	-0.009	-0.003	0.002
	(0000)	(0.019)	(0.079)	(0.851)	(0.485)	(0.849)	(0000)	(0.001)	(0.236)	(0.891)	(0.962)	(0.980)
Sector	0.066	0.049	0.023	0.086	0.095	0.134*	0.051	0.005	-0.008	-0.095	0.017	0.069
	(0.250)	(0.414)	(0.703)	(0.166)	(0.178)	(0.056)	(0.337)	(0.941)	(0.903)	(0.147)	(0.801)	(0.331)
DIM	0.261***	0.344***	0.237***	0.174***	0.182**	0.273***	0.272***	-0.045	-0.083	-0.035	0.015	-0.164^{**}
	(0000)	(0000)	(0000)	(0.014)	(0.011)	(000.0)	(0000)	(0.496)	(0.222)	(0.619)	(0.827)	(0.027)
GDP [%]	0.031	0.087	0.251***	0.116*	0.051	0.261***	-0.141***	-0.156^{**}	—0.142**	-0.152**	-0.202^{***}	-0.164^{**}
	(0.612)	(0.148)	(0000)	(0.082)	(0.440)	(000.0)	(600.0)	(0.012)	(0.027)	(0.021)	(0.003)	(0.021)
R ²	0.249	0.236	0.195	0.195	0.103	0.179	0.361	0.144	0.079	0.056	0.098	0.098
ш	6.811	6.264	4.790	4.554	2.000	3.442	11.600	3.438	1.758	1.188	1.972	1.803
As in Table 3. Source: The Author.												

3048 🕳 T. DUDYCZ

Table 7. Regression	analysis of th	ie CAR and BH	AR. Independer	it variables froi	m the year befo	ore IPO.				
						BHAR	BHAR	BHAR	BHAR	BHAR
	CAR IPO+1	CAR IPO+2	CAR IPO+3	CAR IPO+4	CAR IPO+5	IPO+1	IPO+2	IPO+3	IPO+4	IPO+5
Panel A: Mean value										
Z	259	259	258	247	232	259	259	258	247	232
Mean	0.293	0.271	0.237	0.254	0.251	0.125	0.039	-0.033	-0.065	-0.105
Trimmed mean 5%	0.039	0.010	-0.018	-0.061	-0.060	-0.018	-0.088	-0.176	-0.271	-0.329
SD	3.378	3.348	3.429	3.757	3.896	1.143	1.058	1.284	1.620	1.596
Median	0.039	-0.035	-0.030	-0.028	-0.067	-0.072	-0.194	-0.297	-0.394	-0.470
Panel B: Regression anal	ysis									
ROE_1	0.020	0.006	0.009	0.019	0.011	0.165**	0.138**	0.138*	0.114	0.092
	(0.774)	(0.933)	(0.895)	(0.791)	(0.877)	(0.017)	(0.049)	(0.050)	(0.114)	(0.212)
S/A_1	0.014	0.020	0.026	0.024	0.021	-0.053	-0.054	-0.044	-0.032	-0.052
	(0.828)	(0.757)	(0.687)	(0.719)	(0.763)	(0.411)	(0.410)	(0.507)	(0.632)	(0.450)
CR_{-1}	-0.077	-0.101	-0.100	-0.102	-0.115	-0.021	0.006	0.010	-0.021	-0.026
	(0.276)	(0.153)	(0.154)	(0.159)	(0.124)	(0.770)	(0.932)	(0.892)	(0.779)	(0.732)
D/A_{-1}	-0.050	-0.046	-0.058	-0.058	-0.056	-0.010	0.014	-0.008	090.0	0.072
	(0.475)	(0.508)	(0.411)	(0.415)	(0.448)	(0.891)	(0.847)	(0.911)	(0.408)	(0.334)
E/FA_1	0.009	0.022	0.026	0.019	0.031	0.014	0.001	0.008	0.020	0.024
	(0.898)	(0.745)	(0.709)	(0.781)	(0.667)	(0.835)	(0.986)	(0.908)	(0.776)	(0.738)
SC/E_1	-0.022	-0.024	-0.031	-0.015	-0.004	0.155**	0.125*	0.025	-0.044	-0.017
	(0.744)	(0.720)	(0.653)	(0.839)	(0.961)	(0.023)	(0.070)	(0.713)	(0.552)	(0.823)
$Ln S_{-1}$	-0.145^{*}	-0.148^{*}	-0.155^{**}	-0.166^{**}	-0.183^{**}	-0.027	0.070	0.072	0.019	0.036
	(0.057)	(0.051)	(0.041)	(0.034)	(0.023)	(0.718)	(0.359)	(0.346)	(0.813)	(0.651)
Ln NrSh	0.080	0.091	0.103	0.074	0.066	0.028	0.042	0.057	0.005	-0.067
	(0.258)	(0.197)	(0.143)	(0.308)	(0.366)	(0690)	(0.552)	(0.426)	(0.946)	(0.366)
IP/NP	-0.041	-0.045	-0.053	-0.063	-0.062	-0.033	-0.033	-0.055	-0.066	-0.052
	(0.539)	(0.504)	(0.426)	(0.358)	(0.381)	(0.617)	(0.623)	(0.418)	(0.343)	(0.466)
Sector	-0.063	-0.059	-0.065	-0.072	-0.064	-0.040	0.034	-0.022	-0.004	0.052
	(0.328)	(0.363)	(0.314)	(0.284)	(0.353)	(0.528)	(0.602)	(0.737)	(0.952)	(0.453)
WIG 0	0.060	0.051	0.059	0.052	0.029	0.047	-0.041	0.010	—0.042	0.005
	(0.387)	(0.459)	(0.390)	(0.455)	(0.688)	(0.494)	(0.551)	(0.882)	(0.557)	(0:950)
GDP _1 [%]	-0.078	-0.066	-0.082	-0.084	-0.113	-0.061	-0.030	-0.032	-0.046	-0.126^{*}
	(0.229)	(0.309)	(0.207)	(0.204)	(0.099)	(0.350)	(0.652)	(0.627)	(0.498)	(0.068)
\mathbb{R}^2	0.046	0.047	0.058	0.056	0.064	0.057	0.037	0.028	0.028	0.046
ц	1.010	1.017	1.258	1.175	1.255	1.262	0.803	0.584	0.555	0.872
Panel A reports the me	an values of eac	h variable Y _r . IPO-	1 to IPO + 5 den	ote the year in re	lation to IPO, whe	re IPO $+$ 0 is th	e year of IPO.			
The results of the regre	ssions from the	model II _M are rep	orted in Panel B.	577 - VP - VP	-		-		-	4
The constant term is n	ot reported. <i>P</i> -va	alues are reported	I in parentheses.	(*), (**) and (***) indicate that co	efficients are sig	jnificant at the	10, 5 and 1 per	rcent levels of s	ignificance,
respecuvery. Courses The Author										
Source: IIIE Autiloi.										

new issue, i.e., on IP/BV, but in the year of issue and the subsequent years after the issue, no impact on MV/BV can be observed, except for one year in which it is negative and one year in which it is positive. However, it can also be seen that in addition to SC/E, IP/BV correlates with profitability (ROE), leverage (D/A) - which shows positive verification by banks, low supply of shares (Ln NrSh), IP/NP, the inverse of which shows what part of the IP will be booked as share capital and will affect the share of the new issue in the share capital, and timing (WIG). A positive correlation between these parameters and IP/BV seems justified because each of them (except IP/ NP) can be a positive signal for investors. However, in the years following the debut, except for timing (WIG), the impact of these parameters on MV/BV almost disappears. This suggests that when a company goes public, it is a new entity on the stock exchange and is not known to investors. It publishes a prospectus with extensive information, including accounting data. Investors absorb this information and it influences their decisions. Thus, it can be said that in this case, there is information efficiency in the market. However, after the IPO, the situation changes. The public company is now just one of the many securities. Investors make their decisions in blocks (Windolf, 2016), counting mainly on the short-term benefit (Kim et al., 2017), which has already been called short-termism (Tonello, 2006). Stock prices break away from the fundamentals, and even such absurd situations as the negative impact of profitability on valuation can be observed (Table 5 Panel B1). The only parameter that maintains a constant impact on the valuation is WIG, which means that investor optimism, not accounting information, is crucial to the valuation of shares. In this context, one can also explain the disappearance of the positive impact of SC/E on MV/BV in the years following a company's listing on the stock exchange. Therefore, the utility of SC as a signalling tool is limited to a company's debut only and disappears in the years after its debut. However, if we assume that the reason for the disappearance is low information efficiency of the market, then the obtained results tend to confirm H1. In a broader context, this also means that the information efficiency of the market is only noticeable when a company goes public, while it is very poor or even disappears after the IPO.

4.3. The predictive power of pre-IPO accounting information on a company's performance after IPO

Accounting information before an IPO is published in prospectuses and usually comes from the financial statements at the end of the year preceding the IPO. In general, previous studies have shown that the time of debut does not affect the financial results of companies going public (Dudycz & Brycz, 2017). However, financial parameters that correlated with pre-IPO performance have very little predictive power for accounting performance after an IPO. As shown in Tables 3 and 4 Panel B2, CR has relatively the strongest prediction power, which negatively affects both ROA and ROE three years after a company's debut. Hence the conclusion that companies that had excessive cash levels before their debuts cannot cope with investing effectively after their debuts as well. Another parameter with very short-term predictive ability is SC/ E, which shows a relationship with ROA in the two immediate years after an IPO

Instruments: ROS, S/A, CR, D/A, E/FA, NP, SGrowth, Ln	Assets, Sector, WIG, GDP [%]	
	ROA _{IPO-1}	ROE _{IPO-1}
ROS	0.000	0.000
	(0.723)	(0.951)
S/A	0.002	0.003
	(0.218)	(0.270)
CR	-0.005	-0.012
	(0.158)	(0.072)*
D/A	-0.034	0.184
	(0.356)	(0.010)***
E/FA	0.001	0.002
	(0.578)	(0.361)
SC/E	-0.121	-0.247
	(0.089)*	(0.072)*
Ln Assets	-0.037	-0.060
	(0.000)***	(0.001)***
Sector	0.002	0.004
	(0.387)	(0.383)
WIG	0.000	0.000
	(0.204)	(0.359)
GDP [%]	0.003	0.004
	(0.504)	(0.663)
R ²	0.119	0.131
F	4.066	4.439
Hausman's test	1.997	2,230
	(0.158)	(0,135)
Sargan test	1.662	0,974
-	(0.197)	(0,324)
Weak instrument test: First-stage - statistic	F (2. 247	7) = 5.275
The table reports the SCLC results. For independent yes	ishlar, the values of the "h" coeffici	ionts of the model are pro

Table	8.	2SLS	regressions	examining	the	effect	of	share	capital	on	the	accounting	perform-
ance n	nea	sures.											

The table reports the 2SLS results. For independent variables, the values of the "b" coefficients of the model are presented. *P*-values are reported in parentheses. (*), (**) and (***) indicate that coefficients are significant at the 10, 5 and 1 per cent levels of significance, respectively.

Source: The Author.

Parameter modified by instruments: SC/E

and with ROE in one year. In turn, the size of the company (Ln Assets) is related to ROA and ROE only in the year of the IPO. The leverage behaviour is quite specific. Its high level positively affects ROE before an IPO, but due to the decrease in ROA after the IPO and the occurrence of a negative leverage effect (Schwarz, 2018), the relationship with ROE turns negative.

But what is more important, the results from before an IPO do not predict well the market achievements of companies (Tables 5 and 6 Panel B2). In the case of MV/ BV, they are basically related only to the results achieved in the IPO year. It should be noted, however, that IP/BV is the issue price paid before IPO to book value and $MV/BV_{IPO + 0}$ is the market value at the end of the IPO year to book value after posting the new issue. The time difference may be small – even about a month, so investors basically value the company based on information from prospectuses, as the new financial statement has not yet been published. We can say there is some prognostic power in the case of SC/E, which shows a relationship with MV/BV three years after the IPO and Q-T in two. This justifies the hypothesis that SC can be a signalling tool. IP/NP, whose high value reflects perceptual biases committed by investors, shows little prognostic power. However, there is a clear connection between timing,

	•		-									
	MV/BV _{IPO+0}	MV/BV _{IPO+1}	MV/BV _{IPO+2}	MV/BV _{IPO+3}	MV/BV_{IPO+4}	MV/BV _{IPO+5}	Q-T IPO+0	Q-T _{IPO+1}	Q-T IPO+2	Q-T IPO+3	Q-T _{PO+4}	Q-T IPO+5
ROE	0.236***	-0.008	0.113	-0.166^{**}	0.548***	0.197	0.218***	0.163**	0.143*	-0.014	-0.723^{***}	-0.080
	(0000)	(0.912)	(0.129)	(0.038)	(0000)	(0.235)	(0000)	(0.019)	(0.058)	(0.861)	(0000)	(0.646)
S/A	-0.018	-0.056	0.021	0.028	-0.028	0.017	-0.025	0.039	0.050	0.050	0.017	-0.085
	(0.738)	(0.375)	(0.750)	(0.678)	(0.679)	(0.847)	(0.642)	(0.519)	(0.442)	(0.455)	(0.776)	(0.341)
ß	0.100*	0.096	-0.084	-0.114	-0.042	-0.016	0.068	-0.129^{*}	-0.109	-0.081	-0.087	-0.024
	(0.092)	(0.163)	(0.224)	(0.135)	(0.515)	(0.835)	(0.241)	(0.053)	(0.118)	(0.283)	(0.117)	(0.765)
D/A	0.129**	-0.017	0.107	0.064	0.173**	0.100	-0.117**	-0.077	0.006	-0.060	-0.047	0.093
	(0.028)	(0.801)	(0.126)	(0.396)	(0.017)	(0.201)	(0.043)	(0.226)	(0:930)	(0.421)	(0.449)	(0.248)
E/FA	-0.091	0.045	-0.045	-0.034	-0.018	-0.029	-0.122**	-0.047	-0.132^{**}	-0.152^{**}	-0.086^{*}	-0.124^{*}
	(0.114)	(0.477)	(0.463)	(0.602)	(0.767)	(0.687)	(0.031)	(0.442)	(0.034)	(0.018)	(0.099)	(0.098)
SC/E	0.279***	0.109	-0.099	-0.196^{**}	0.174**	0.307*	0.374***	-0.032	-0.047	-0.094	-0.395***	-0.053
	(0000)	(0.134)	(0.197)	(0.021)	(0.012)	(0.065)	(0000)	(0.645)	(0.543)	(0.260)	(0000)	(0.758)
Ln S	-0.020	0.287***	-0.141	-0.236^{***}	-0.088	0.091	-0.001	-0.358^{***}	-0.229^{**}	-0.150^{*}	-0.025	0.058
	(0.747)	(0000)	(0.109)	(00.0)	(0.265)	(0.325)	(0.983)	(000.0)	(0.010)	(0.091)	(0.714)	(0.541)
Ln NrSh	-0.134^{**}	-0.198^{***}	0.021	0.106	-0.051	-0.243***	-0.116^{**}	0.159**	0.061	0.025	-0.007	-0.102
	(0.023)	(0000)	(0.787)	(0.179)	(0.475)	(0.004)	(0.045)	(0.029)	(0.430)	(0.742)	(906.0)	(0.231)
IP/NP	0.217***	0.051	0.008	0.027	-0.023	-0.030	0.253***	0.189**	0.073	0.026	-0.029	-0.043
	(0000)	(0.450)	(0.895)	(0.682)	(0.711)	(0.665)	(0000)	(0.004)	(0.259)	(0.697)	(0.583)	(0.547)
Sector	0.058^{*}	0.079	0.084	0.017	0.006	0.152**	0.051	0.007	0.038	0.102	0.121**	0.111
	(0.282)	(0.212)	(0.190)	(0.806)	(0.924)	(0:039)	(0.337)	(0.906)	(0.561)	(0.132)	(0.025)	(0.145)
DIM	0.284***	-0.047	0.237***	0.049	-0.051	0.150**	0.272***	-0.005	0.156**	0.148**	-0.009	0.004
	(0000)	(0.523)	(0.001)	(0.497)	(0.465)	(0.045)	(0000)	(0.947)	(0.024)	(0.037)	(0.873)	(0.961)
GDP [%]	-0.113^{**}	0.110	-0.175***	0.048	0.113*	0.072	-0.141***	-0.177^{***}	-0.089	0.055	0.060	0.102
	(0.039)	(0.104)	(0.007)	(0.487)	(0.089)	(0.292)	(0.00)	(0.007)	(0.172)	(0.416)	(0.287)	(0.150)
\mathbb{R}^2	0.337	0.094	0.110	0.061	0.242	0.118	0.361	0.164	0.100	0.086	0.448	0.062
ц	10.442	2.136	2.514	1.248	5.560	2.143	11.600	4.003	2.567	1.811	13.986	1.051
The table	reports the re	sults of the re	gression: $Y_t =$	$a + \beta_1 ROE_{t-1} +$	$-\beta_2 \frac{5}{A_{t-1}} + \beta_3 CR$	$_{t-1}^{t}+\ \beta _{4}\frac{\textit{D}}{\textit{A}_{t-1}}\ +$	$\beta_5 \frac{E}{E_A t_{-1}} + \beta_6 \frac{SC}{E_1}$	$-1 + \beta_7 Ln S_{t-1}$	$_{1}$ + $\beta_{8}NrSh_{t-}$	$_{1} + \beta_{9} \frac{l^{p}}{w^{p}} +$	β_{10} Sector + β_{11}	$WIG_{t-1} +$
$\beta_{12}GDP[\%$	$\left[\int_{t-1} + \varepsilon_{t-1} \cdot \mathbf{t} \right]$	the year relativ	ve to IPO (whe	re IPO $+$ 1 is the theorem of the theorem of the tension of tens	ne first year aft	ter debuť). Ťhe	constant term	is not reported	l. P-values are	reported in I	parentheses. ($*$	i, (**) and
(***) indic	ate that coeffic	ients are signific	cant at the 10,	5 and 1 per cen	t levels of sign	ificance, respect	tively.					
Source: Ih	e Author.											

Table 9. Regression analysis of the market performance with lagged explanatory variables.

3052 🍚 T. DUDYCZ

especially with regard to GDP, and market achievements. Interestingly, however, it is a negative relationship and visible only in Q-T. It may suggest that companies that debuted during the downturn actually needed capital to finance their investments and did not just benefit from a good opportunity to raise cheap capital. However, what is very important for investors is that pre-IPO financial results are very poorly useful for forecasting future returns. Table 7 shows that they do not show any relationship with CAR, while with BHAR, there is a relationship with ROE for three years and one year with SC/E. It follows that hypothesis 3 has not been confirmed.

4.5. Robustness checks

4.5.1. Endogeneity

Given the endogenous nature of accounting information, it can be assumed that there is a simultaneous or inverse causality between the ratio of share capital to equity (SC/ E) and accounting performance measures. Firms that perform well accumulate more profit which, booked as reserve capital, increases equity and thus contributes to a decline in SC/E with a constant SC. Also, in the course of the financial year, the earned profit goes to equity. This phenomenon may disrupt the results that we have obtained. In order to alleviate these concerns, we refer to our previous studies, which show that the structure of equity is not only the result of the scale of the previous accumulation of profit, but is also a result of strategies undertaken in this regard (Dudycz & Brycz, 2021). Firms that introduce a small amount of share capital break it down into shares with a low nominal price to ensure that the number of shares is in the desired range, and the existing capital needs are largely supplemented with other types of equity, so SC/E is low. Conversely, firms that inject large amounts of share capital break it down into shares with a higher nominal price and use less equity that is not SC, and thus have a lower SC/E. This is the source of the correlation between the nominal share price (NP) and SC/E. This will be used in the selection of instruments for the two-stage least square regression (2SLS in Gretl), which we used to address the concern of endogeneity (Table 8). The second instrumental variable introduced in the regression of accounting results is sales growth (SGrowth), which is a proxy for IPO-firm quality (following Zheng & Stangeland, 2007). We measured sales growth from one year before the IPO to the IPO year. Three diagnostic tests were used: (1) Hausman's test, to detect endogenous regressors, (2) the Sargan test, for over-identification of applied instrumental variables, and (3) the weak instrument test, to test the quality of the instruments used. Not rejecting the null hypothesis in Hausman's test suggests that the OLS estimator is effective. Failure to reject the null hypothesis in the Sargan test suggests that all instruments in the model are valid. A value for the F statistic of less than 10 means that the instruments are weak in explaining the SC/E regressor, which means that the significance of the parameters in 2SLS is lower than in OLS. Overall, our tests for endogeneity suggest that SC/E is exogenous. However, we need to acknowledge that the weakness of the instruments is a limitation of our study.

4.5.2. Regression with lagged explanatory variables

The strength of the results, indicating that investors absorb information about IPO companies from prospectuses and no longer absorb information from financial statements in the years following the debut, are indicated by the regression results with lagged variables. In Tables 5 and 6, comparing the results for the dependent variable from the IPO + 0 period, we can see that in Panel B both the model fit (R²) and the explanatory power of the regressors (F statistic > 10) are higher than for the data shown in Panel A. Panel B examines the impact of variables from the IPO-1 period obtained from the prospectus, while Panel A examines the impact of variables from the IPO + 0 period obtained from the financial statements of that period. This means that in the year of the debut (IPO + 0), the information from the previous year disclosed in the prospectus has a greater impact on the market valuation than the information from the financial statements of the current year. This may give rise to an assumption that information from the previous year always has a greater impact on the market performance in the next year than the results of a given year. For this purpose, regression models with delayed explanatory variables by one year were constructed (Table 9). As we can see, starting from the IPO + 1 year, the impact of the previous year's results on the company's market performance in the next year is low and very ambiguous. This suggests that, after the debut, both the results disclosed in the financial statements of a given year and the results from the previous year have a very low impact on the market performance of the company, which strengthens the claim that the information efficiency of the capital market is low in the years following the debut.

5. Discussion and conclusions

The purpose of the article was to examine the impact of share capital on companies' performance and, additionally, the impact of accounting information on market performance of companies. as well as the influence of accounting information before an IPO on the predictive power of achievements after an IPO. The research showed that a large amount of share capital in equity can reduce a company's capital flexibility. which is in line with Ho's (2017) suggestion. This conclusion, however, should be limited to companies where there is a need to reduce capital due to their deteriorating accounting results. The studies do not offer grounds for making conclusions about the impact of share capital on financial flexibility in the event of a positive investment shock and an increase in the demand for capital. However, the results of the research provide grounds for stating that share capital can be a tool to signal the degree of shareholder involvement, which translates into better valuation of shares. These results should be taken into account when analysing shareholders' information expectations. Wide popularization and implementation of International Financial Reporting Standards (IFRS) representing the Anglo-American accounting model has resulted in the dissemination of fair value as a common accounting paradigm. However, fair value is a very fragile measure due to the large fluctuation in market prices and is also highly subjective for unobservable assets on the market. Therefore, whether completely depriving shareholders of all permanent and objective information is the correct direction of change in accounting should be considered. All the more so because there are voices saying that the Anglo-American model of corporate governance and financial reporting is dysfunctional and a source of crisis (Sikka & Stittle, 2019). In the context of popularizing IFRS, as well as learning about the mechanisms occurring in the 'relational investor' markets, especially emerging markets, the results showing low market information efficiency after an IPO are important. The more so that, as indicated by Joseph et al. (2020), despite the coordinated economic and financial policies of EU members, the markets of Eastern Europe do not show the desired level of market integration. The article has shown that information efficiency is noticeable during IPOs because of the great interest of investors in new companies, but it disappears after the IPO. Stock prices break away from the fundamentals, and timing is the main determinant of market performance. The implementation and dissemination of IFRS is accompanied by many expectations and hope. The results obtained are relevant to research on the usability of accounting standards. Creating and testing new standards will be ineffective in a situation where the market does not absorb information.

The article also showed that information that correlated with the valuation of shares at the time of an IPO has little effect on the predictive power of companies' performance after the IPO. This is another confirmation of the low information efficiency of the market. Since, as suggested earlier, post-IPO share prices break away from the fundamentals, they may not therefore be related to pre-IPO information. One of the reasons for this state of affairs may be seen in the structural changes of investors and the resulting investment behaviour. According to Sikka and Stittle (2019), the number of shares held by individual investors steadily decreases. Parallel to the falling share capital held by individual investors, the turnover rate increases. For example, for the 1,000 largest companies in the US, it increased from 12% in 1960 to more than 100% in 2002 (Windolf, 2016). The share of foreign investors in the stock markets is growing, for example, in Poland belong to them about 60% of turnover. The accompanying language barrier may hamper access to all information. Short-termism is becoming the dominant investment strategy. This strategy does not need information on the foundations of companies. It deals only with tracking moods. Stock prices are not driven by expectations but by expected expectations. Expected expectations replace information (Windolf, 2016). These phenomena may appear faster and more strongly in small and underdeveloped capital markets. This article brings the knowledge for practitioners, investors and policymakers trying to understand what influences the performance of newly listed firms. Accounting information does not captures long-term consequences and market performance is detached from the fundamentals presented in the financial statements. As a result, achievements are largely created outside companies by external actors. In this situation, the influence of the management board on creating value for shareholders is limited. In this context, expectations that the new accounting standards will contribute to an increase in the allocation efficiency of capital markets may be futile.

Disclosure statement

No potential conflict of interest was reported by the authors.

Funding

This work was supported by the National Science Centre Poland [grant number: 2017/25/B/ HS4/01374].

ORCID

Tadeusz Dudycz i http://orcid.org/0000-0002-0409-0061

References

- Ajekwe, C. C. M., & Ibiamke, A. (2018). Financial statement analysis to predict stock returns of listed consumer goods firms in Nigeria. *Journal of Finance and Accounting*, 6(1), 27–33.
- Alexakis, C., Patra, T., & Poshakwale, S. (2010). Predictability of stock returns using financial statement information: Evidence on semi-strong efficiency of emerging greek stock market. *Applied Financial Economics*, 20(16), 1321–1326. https://doi.org/10.1080/09603107.2010. 482517
- Aluchna, M., & Kaminski, B. (2017). Ownership structure and company performance: A panel study from Poland. *Baltic Journal of Management*, 12(4), 485–502. https://doi.org/10.1108/ BJM-01-2017-0025
- Amel-Zadeh, A., Calliessz, J.-P., Kaiserx, D., & Roberts, S. (2020). Machine learning-based financial statement analysis. Working paper; January 15, 2020. https://ssrn.com/abstract= 3520684
- Ardalan, K. (2017). Capital structure theory: Reconsidered. Research in International Business and Finance, 39, 696–710. https://doi.org/10.1016/j.ribaf.2015.11.010
- Armour, J. (2000). Share capital and creditor protection: Efficient rules for a modern company law. *The Modern Law Review*, 63(3), 355–378. https://doi.org/10.1111/1468-2230.00268
- Auret, C. J., & Britten, J. H. C. (2008). Post-issue operating performance of firms listing on the JSE. *Investment Analysts Journal*, 37(68), 21–29. https://doi.org/10.1080/10293523.2008. 11082501
- Bachner, T. (2009). Creditor protection in private companies: Anglo-German perspectives for a European legal discourse. Cambridge University Press.
- Ball, R. (2006). International financial reporting standards (IFRS): Pros and cons for investors. Accounting and Business Research, 36(sup1), 5–27. https://doi.org/10.1080/00014788.2006. 9730040
- Ball, R., Kothari, S. P., & Robin, A. (2000). The effect of international institutional factors on properties of accounting earnings. *Journal of Accounting and Economics*, 29(1), 1–51. https://doi.org/10.1016/S0165-4101(00)00012-4
- Bancel, F., & Mittoo, U. R. (2004). Cross-country determinants of capital structure choice: A survey of European firms. *Financial Management*, 33, 103–132.
- Bebchuk, L. A. (1992). Federalism and the corporation: The desirable limits on state competition in corporate law. *Harvard Law Review*, 105(7), 1435–1510. https://doi.org/10.2307/1341744
- Bhabra, H. S., & Pettway, R. H. (2003). IPO prospectus information and subsequent performance. *The Financial Review*, 38(3), 369–397. https://doi.org/10.1111/1540-6288.00051
- Bonbright, J. C. (1924). The dangers of shares without par value. *Columbia Law Review*, 24(5), 449-468.
- Boubakri, N., & Ghouma, H. (2010). Control/ownership structure, creditor rights protection, and the cost of debt financing: International evidence. *Journal of Banking & Finance*, 34(10), 2481–2499. https://doi.org/10.1016/j.jbankfin.2010.04.006
- Brounen, D., de Jong, A., & Koedijk, K. (2006). Capital structure policies in Europe: Survey evidence. *Journal of Banking & Finance*, 30(5), 1409–1442. https://doi.org/10.1016/j.jbankfin. 2005.02.010

- Cai, J., & Wei, K. J. (1997). The investment and operating performance of Japanese initial public offerings. *Pacific-Basin Finance Journal*, 5(4), 389–417. https://doi.org/10.1016/S0927-538X(97)00021-8
- Cascino, S., Clatworthy, M., Osma, B. G., Gassen, J., Imam, S., & Jeanjean, T. (2013). *The use of information by capital providers. EFRAG-ICAS Academic literature review*. http://www.ifrs. org/Meetings/MeetingDocs/IASB/2014/January/AP13-

The use of information by capital providers.pdf.

- Chen, P., & Zhang, G. (2007). How do accounting variables explain stock price movements? Theory and evidence. *Journal of Accounting and Economics*, 43(2–3), 219–244. https://doi.org/10.1016/j.jacceco.2007.01.001
- Cho, S.-S., Ghoul, S. E., Guedhami, O., & Suh, J. (2014). Creditor rights and capital structure: Evidence from international data. *Journal of Corporate Finance*, 25, 40–60. https://doi.org/ 10.1016/j.jcorpfin.2013.10.007
- Chung, H. Y., & Kim, J.-B. (2001). A structured financial statement analysis and the direct prediction of stock prices in Korea. *Asia-Pacific Financial Markets*, 8(2), 87–117. https://doi.org/10.1023/A:1011951102564
- Chung, K. H., & Pruitt, S. W. (1994). A simple approximation of Tobin's q. Financial Management, 23(3), 70-74. https://doi.org/10.2307/3665623
- Cook, W. W. (1921). Stock without par value. American Bar Association Journal, 7(10), 534–537.
- Danielson, M. G., & Press, E. (2003). Accounting returns revisited: Evidence of their usefulness in estimating economic returns. *Review of Accounting Studies*, 8(4), 493–530. https://doi.org/ 10.1023/A:1027368116754
- DeAngelo, H., DeAngelo, L., & Whited, T. M. (2011). Capital structure dynamics and transitory debt. *Journal of Financial Economics*, 99(2), 235-261. https://doi.org/10.1016/j.jfineco. 2010.09.005
- Dudycz, T., & Brycz, B. (2017). What drives the amount of capital raised at IPO? Evidence from the Warsaw Stock Exchange. *Argumenta Oeconomica*, 2(39), 61-89.
- Dudycz, T., & Brycz, B. (2021). Why the par value of share matters to investors. *International Journal of Financial Studies*, 9(1), 16. https://doi.org/10.3390/ijfs9010016
- Enriques, L., & Macey, J. R. (2001). Creditors versus capital formation: The case against the European legal capital rules. *Cornell Law Review*, 86(6), 1165–1204.
- Espenlaub, S., Goyal, A., & Mohamed, A. (2020). The impact of shareholders and creditors rights on IPO performance: An international study. *The British Accounting Review*, 52(1), 100872. https://doi.org/10.1016/j.bar.2019.100872
- Frank, K. (2002). The effect of growth on the value relevance of accounting data. Journal of Business Research, 55(1), 69-78. https://doi.org/10.1016/S0148-2963(00)00131-4
- Goslin, J., Chai, D., & Gunasekarage, A. (2012). The usefulness of financial statement information in predicting stock returns: New Zealand evidence. *Accounting, Business and Finance Journal*, 6(2), 51–70.
- Graham, J. R., & Harvey, C. R. (2001). The theory and practice of corporate finance: Evidence from the field. *Journal of Financial Economics*, 60(2–3), 187–243. https://doi.org/10.1016/S0304-405X(01)00044-7
- Handschin, L. (2012). Risk-based equity requirements: How equity rules for the financial sector can be applied to the real economy. *Journal of Corporate Law Studies*, 12(2), 255–293. https://doi.org/10.5235/JCLS.12.2.255
- Ho, J. K. S. (2017). Revisiting the relevance of the legal capital regime in modern company law. *The Journal of Comparative Law*, 12(1), 1–21.
- Ho, Y. K., & Lan, L. L. (1999). The par value of shares: An irrelevant concept in modern company law. *Singapore Journal of Legal Studies*, 552–572.
- Jain, B. A., & Kini, O. (1994). The post-issue operating performance of IPO firms. *The Journal of Finance*, 49(5), 1699–1726. https://doi.org/10.1111/j.1540-6261.1994.tb04778.x
- Joos, P., & Lang, M. (1994). The effects of accounting diversity: Evidence from the European Union. *Journal of Accounting Research*, *32*, 141––168. https://doi.org/10.2307/2491444

- Joseph, N. L., Vo, T. T. A., Mobarek, A., & Mollah, S. (2020). Volatility and asymmetric dependence in Central and East European stock markets. *Review of Quantitative Finance and Accounting*, 55(4), 1241–1303. https://doi.org/10.1007/s11156-020-00874-0
- Kapopoulos, P., & Lazaretou, S. (2007). Corporate ownership structure and firm performance: Evidence from Greek firms. *Corporate Governance: An International Review*, 15(2), 144–158. https://doi.org/10.1111/j.1467-8683.2007.00551.x
- Kaserer, C., & Moldenhauer, B. (2008). Insider ownership and corporate performance: Evidence from Germany. *Review of Managerial Science*, 2(1), 1–35. https://doi.org/10.1007/s11846-007-0009-3
- Kim, K. A., Kitsabunnarat, P., & Nofsinger, J. R. (2004). Ownership and operating performance in an emerging market: Evidence from Thai IPO firms. *Journal of Corporate Finance*, 10(3), 355–381. https://doi.org/10.1016/S0929-1199(02)00019-6
- Kim, Y., Su, L. N., & Zhu, X. K. (2017). Does the cessation of quarterly earnings guidance reduce investors' short-termism? *Review of Accounting Studies*, 22(2), 715–752. https://doi. org/10.1007/s11142-017-9397-z
- Kothari, S. P. (2001). Capital markets research in accounting. Journal of Accounting and Economics, 31(1-3), 105-231. https://doi.org/10.1016/S0165-4101(01)00030-1
- Kourtis, G. P., Kourtis, E. P., Kourtis, M. P., & Curtis, P. G. (2017). Fundamental analysis, stock returns and high B/M companies. *International Journal of Economics and Business Administration*, V(4), 3–18. https://doi.org/10.35808/ijeba/139
- KPMG. (2008). Feasibility study on an alternative to the capital maintenance regime established by the Second Company Law Directive 77/91/EEC of 13 December 1976 and an examination of the impact on profit distribution of the new EU-accounting regime. KPMG.
- Krivogorsky, V., & Grudnitski, G. (2010). Country-specific institutional effects on ownership: Concentration and performance of continental European firms. *Journal of Management & Governance*, 14(2), 167–193. https://doi.org/10.1007/s10997-009-9097-6
- Lambrinoudakis, C., Skiadopoulos, G., & Gkionis, K. (2019). Capital structure and financial flexibility: Expectations of future shocks. *Journal of Banking & Finance*, 104, 1-18. https://doi.org/10.1016/j.jbankfin.2019.03.016
- Le, T. P. V., & Phan, T. B. N. (2017). Capital structure and firm performance: Empirical evidence from a small transition country. *Research in International Business and Finance*, 42, 710–726. https://doi.org/10.1016/j.ribaf.2017.07.012
- Maury, B. (2006). Family ownership and firm performance: Empirical evidence from Western European corporations. *Journal of Corporate Finance*, *12*(2), 321–341. https://doi.org/10. 1016/j.jcorpfin.2005.02.002
- Mikkelson, W. H., Partch, M. M., & Shah, K. (1997). Ownership and operating performance of companies that go public. *Journal of Financial Economics*, 44(3), 281–307. https://doi.org/ 10.1016/S0304-405X(97)00006-8
- Modigliani, F., & Miller, M. (1958). The cost of capital, corporation finance, and the theory of investment. *The American Economic Review*, 48, 655–669.
- Morck, R., Yeung, B., & Yu, W. (2000). The information content of stock markets: Why do emerging markets have synchronous stock price movements? *Journal of Financial Economics*, 58(1-2), 215-260. https://doi.org/10.1016/S0304-405X(00)00071-4
- Mulbert, P. O., & Birke, M. (2002). Legal capital Is there a case against the European legal capital rules? *European Business Organization Law Review*, 3(4), 695–732. https://doi.org/10. 1017/S1566752900001129
- Mun, S. G., & Jang, S. C. (2019). Restaurant firms' IPO motivations and post-IPO performances: Staying public, being delisted or merged? *International Journal of Contemporary Hospitality Management*, 31(9), 3484–3502. https://doi.org/10.1108/IJCHM-08-2018-0677
- Mwenda, K. K. (1999). Policy issues underpinning the concept of par value: A comparative legal study. *Mountbatten Journal of Legal Studies*, 3(1), 80–90.
- Niu, X. (2008). Theoretical and practical review of capital structure and its determinants. *International Journal of Business and Management*, 3(3), 133–139.

- Ong, C. Z., Mohd-Rashid, R., & Taufil-Mohd, K. N. (2020). Do institutional investors drive the IPO valuation? *Borsa Istanbul Review*, 20(4), 307–332. https://doi.org/10.1016/j.bir.2020.05.003
- Ou, J. A., & Penman, S. H. (1989). Financial statement analysis and the prediction of stock returns. *Journal of Accounting and Economics*, 11(4), 295–329. https://doi.org/10.1016/0165-4101(89)90017-7
- Pagano, M., Panetta, F., & Zingales, L. (1998). Why do companies go public? An empirical analysis. *The Journal of Finance*, 53(1), 27–64. https://doi.org/10.1111/0022-1082.25448
- Panetsos, L. (2016). Accounting standards and legal capital in EU law. Utrecht Law Review, 12(1), 139–158. https://doi.org/10.18352/ulr.322
- Paniagua, J., Rivelles, R., & Sapena, J. (2018). Corporate governance and financial performance: The role of ownership and board structure. *Journal of Business Research*, 89, 229–234. https://doi.org/10.1016/j.jbusres.2018.01.060
- Pennington, R. (1990). Pennington's company law. Butterworths.
- Piotroski, J. D. (2000). Value investing: The use of historical financial statement information to separate winners from losers. *Journal of Accounting Research*, 38, 1–41. https://doi.org/10. 2307/2672906
- Pistor, K. (2000). Patterns of legal change: Shareholder and creditor rights in transition economies. European Business Organization Law Review, 1(1), 59–110. https://doi.org/10.1017/ S1566752900000069
- Rickford, J. (2004). Reforming capital report of the Interdisciplinary Group on Capital Maintenance. European Business Law Review, 15, 920–1027.
- Roe, M. J. (1993). Some differences in corporate structure in Germany, Japan, and the United States. *The Yale Law Journal*, 102(8), 1927–2003. https://doi.org/10.2307/796856
- Schwarz, D. (2018). The ability of listed companies to optimize their capital structure, shape their distribution policy and fight hostile takeovers by repurchasing their own shares. *Entrepreneurship and Sustainability Issues*, 6(2), 636–648. https://doi.org/10.9770/jesi.2018.6.2(12)
- Short, H., & Keasey, K. (1999). Managerial ownership and the performance of firms: Evidence from the UK. *Journal of Corporate Finance*, 5(1), 79–101. https://doi.org/10.1016/S0929-1199(98)00016-9
- Sikka, P., & Stittle, J. (2019). Debunking the myth of shareholder ownership of companies: Some implications for corporate governance and financial reporting. *Critical Perspectives on Accounting*, 63, 101992. https://doi.org/10.1016/j.cpa.2017.03.011
- Skogsvik, S. (2008). Financial statement information, the prediction of book return on owners' equity and market efficiency: The Swedish case. *Journal of Business Finance & Accounting*, 35(7-8), 795–817. https://doi.org/10.1111/j.1468-5957.2008.02099.x
- Skogsvik, S., & Skogsvik, K. (2010). Accounting-based probabilistic prediction of ROE, the residual income valuation model and the assessment of mispricing in the Swedish Stock Market. Abacus, 46(4), 387–418. https://doi.org/10.1111/j.1467-6281.2010.00325.x
- Spence, M. (1973). Job market signalling. The Quarterly Journal of Economics, 87(3), 355-374. https://doi.org/10.2307/1882010
- Spence, M. (2002). Signaling in retrospect and the informational structure of markets. *American Economic Review*, 92(3), 434–459. https://doi.org/10.1257/00028280260136200
- Tonello, M. (2006). *Revisiting stock market short-termism*. (Report Number: R-1 386-06-RR). New York: Conference Board.
- Windolf, P. (2016). Riding the bubble: Financial market crises in twenty-two OECD countries. *Journal of Economic Issues*, 50(3), 788–813. https://doi.org/10.1080/00213624.2016.1213588
- Woodley, M. K., Jones, S. T., & Reburn, J. P. (2011). Value stocks and accounting screens: Has a good rule gone bad? *Journal of Accounting and Finance*, 11(4), 87–104.
- Yan, X. S., & Zheng, L. (2017). Fundamental analysis and the cross-section of stock returns: A data-mining approach. *The Review of Financial Studies*, 30(4), 1382–1423. https://doi.org/10. 1093/rfs/hhx001
- Zheng, S., & Stangeland, D. (2007). IPO underpricing, firm quality and analyst forecasts. *Financial Management*, 36(2), 1–64. https://doi.org/10.1111/j.1755-053X.2007.tb00086.x